Author’s response to reviews

Title: Factors related to subjective satisfaction following microendoscopic foraminotomy for cervical radiculopathy

Authors:

Juichi Tonosu (juichitohnosu@yahoo.co.jp)
Hirohiko Inanami (inanamihiro@gmail.com)
Hiroyuki Oka (okah-tky@umin.ac.jp)
Yuichi Takano (luigi.igiul1030@gmail.com)
Hisashi Koga (hkoga0808@gmail.com)
Yohei Yuzawa (heall_p@yahoo.co.jp)
Ryutaro Shiboi (obedifrothblower78@gmail.com)
Yasushi Oshima (yoo-ty@umin.ac.jp)
Satoshi Baba (bababasatoshi0518@yahoo.co.jp)
Sakae Tanaka (tanakas-ort@h.u-tokyo.ac.jp)
Ko Matsudaira (kohart801@gmail.com)

Version: 4 Date: 15 Jan 2018

Author’s response to reviews:

Response to Sondos Majeed’s Comments:

Thank you very much for your valuable recommendation. We have provided a response to your comment below.

Comment. Please note that the trial registration section of the abstract refers to clinical trials. A clinical trial is a research study that prospectively assigns human subjects to one or more health related interventions to evaluate the effects on health or biological outcomes.

As your study does not fall under the category of a clinical trial, and as you have also indicated that your study is not a clinical trial on the questionnaire responses in the submission system, please remove the 'Trial registration' section of the abstract.
Response: As you pointed out, our study does not fall under the category of a clinical trial. Therefore, we have removed the ‘Trial registration’ section of the abstract.